Author’s response to reviews

Title: "How do we use the time?" – an observational study measuring the task time distribution of nurses in psychiatric care

Authors:
Andreas Glantz (andreas.glantz@skane.se)
Karin Örmon (karin.ormon@mah.se)
Boel Sandström (boel.sandstrom@bth.se)

Version: 2 Date: 31 Jul 2019

Author’s response to reviews:

Response to editor and reviewers

Dear Editor-in-Chief and Reviewers,

Thank you for your comments and this opportunity to improve our manuscript. We are happy to provide you with a revised manuscript. We have thoroughly edited the entire manuscript in order to improve flow and style.

The introduction has been updated according to your suggestions.

Line 32, page 2. Line 31-34, page 2. Updated with each benefit together with the appropriate reference.

Line 44, 46, page 3. Revised the entire manuscript to use the same wording everywhere.